Author’s response to reviews

Title: Mapping Evidence on Women’s Knowledge and Practice of Breast Self-Examination in Sub-Saharan Africa: A Scoping Review Protocol

Authors:

Roseline Udoh (henryrose.ru@gmail.com)
Monica Ansu-Mensah (manvam3@yahoo.com)
Mohammed Tahiru (mohammedtahiru66@yahoo.com)
Vitalis Bawontuo (bawontuovitalis@yahoo.com)
Desmond Kuupiel (desmondkuupiel98@hotmail.com)

Version: 2 Date: 28 Nov 2019

Author’s response to reviews:

Please, we have included this as part of the online submission.